Author's response to reviews

Title: Acupuncture for pain and osteoarthritis of the knee: a pilot study for an open parallel-arm randomised controlled trial

Authors:

Harriet Lansdown (harrietlansdown@gmail.com)
Katie Howard (katedrawoh@hotmail.co.uk)
Stephen Brealey (sb143@york.ac.uk)
Hugh MacPherson (hm18@york.ac.uk)

Version: 6 Date: 8 October 2009

Author's response to reviews:

I have changed spacing in Fig 1. Wide spacing had hidden some of the figure in the boxes meaning that the numbers did not add up. Fig 1 renamed 'Fig1 revised 8.10'

Harriet Lansdown
8th Oct 2009